Author's response to reviews

Title: Anti-inflammatory and anti-oxidative effects of corilagin in a rat model of acute cholestasis

Authors:

Feng Jin (jinfengsdjn@sina.com)
Du Cheng (476687947@qq.com)
Jun-Yan Tao (Junyan.Tao2@ucsf.edu)
Ran Pang (pangran05@163.com)
Yuan-Jin Guo (sophiaguo1028@yahoo.com.cn)
Pian Ye (122454993@qq.com)
Ji-Hua Dong (djh7045@126.com)
Shu-Ling Zhang (chineseomd@yahoo.com)
Lei Zhao (chineseemd@hotmail.com)

Version: 4 Date: 3 March 2013

Author's response to reviews: see over
Dear editor,

According to the Associate Editor’s comment, we revised the manuscript carefully. The revisions in detail are listed as follows.

Associate Editor’s Comments:

The authors should re-check figures and figure legends (which are very confusing regarding figure numbers). Moreover the figure legends should be expanded. A reader should be able to understand, what is shown, without reading the whole manuscript!

Response: We corrected the figure numbers and expanded the figure legends so that a reader could understand the content without reading whole article.

If you have any further requirement, please let us know. Thank you very much.

Lei Zhao, M.D., Ph.D., Associate Professor of Medicine
Department of Infectious Disease & Hepatology
Union Hospital
Tongji Medical College
Huazhong University of Science and Technology
Wuhan, 430022, P.R. China
Tel: +86-18971397498
Fax: +86-27-87716917
E-mail: chinesemd@hotmail.com, dr.leizhao@gmail.com
Past contact:
Postdoctoral Research Fellow
GI Unit, Department of Medicine
Massachusetts General Hospital
Harvard Medical School
55 Fruit Street, Boston, MA 02114
Tel: 1-781-354-0485